Author’s response to reviews

Title: A 44-year-old patient with a new-onset seizure disorder after vaccination against Japanese Encephalitis: a case report

Authors:

Christoph Schulze (chrisinro@aol.com)

Version: 5 Date: 6 December 2012

Author’s response to reviews: see over
Rostock, December 6th 2012

Dear Editor in chief,

I enclose the revised manuscript with the title: “A 44-year-old patient with a new-onset seizure disorder after vaccination against Japanese Encephalitis: a case report” as a case report for consideration in the Journal of medical case reports. I thank the editor for the helpful comment. I provide ethnicity of the patient as requested. Changes are highlighted in the text. I hope the provided ethnic information is the kind of information you asked for.

I hope the revision is suitable for publication now.

Sincerely,

Christoph Schulze, MD
First author